Author’s response to reviews

Title: Comprehensive Geriatric Assessment Pilot of a Randomized Control Study in a Swedish Acute Hospital: A Feasibility Study

Authors:

Theresa Westgård (theresa.westgard@neuro.gu.se)
Isabelle Ottenvall Hammar (Isabelle.o-h@neuro.gu.se)
Eva Holmgren (Eva.holmgren@neuro.gu.se)
Anna Ehrenberg (aeh@du.se)
Aase Wisten (aase.wisten@gmail.com)
Anne W Ekdahl (anne.w.ekdahl@skane.se)
Synneve Dahlin-Ivanoff (synneve.dahlin-ivanoff@neuro.gu.se)
Katarina Wilhelmson (katarina.wilhelmson@gu.se)

Version: 2 Date: 27 Oct 2017

Author’s response to reviews:

The manuscript has been revised and the track changes that were on the previous manuscript to show the grammatical changes have been removed in this edition per the reviewers request.